Author's response to reviews

Title: "There was a young man, who swallowed a coin..." Pancreatic and biliary obstruction years after a swallowed foreign body retained in a duodenal diverticulum: A case report

Authors:

Ibrahim Ismail (issey20@gmail.com)
David W Mudge (David.Mudge@health.qld.gov.au)

Version: 7 Date: 11 April 2015

Author's response to reviews:

'To our knowledge, this is the first case to describe a duodenal diverticula...' to:
'To our knowledge, this is the first case to describe a duodenal diverticulum.....' has been changed as request.